Author’s response to reviews

Title: Association between 28 single nucleotide polymorphisms and type 2 diabetes mellitus in the Kazakh population: a case-control study

Authors:

Nurgul Sikhayeva (ksnurgul@gmail.com)
Aisha Iskakova (aishaisk1@gmail.com)
Nuria Saigi-Morgui (nuria.saigi-morgui@chuv.ch)
Elena Zholdybayeva (lenazhol@gmail.com)
Chin-Bin Eap (chin.eap@chuv.ch)
Erlan Ramanculov (ramanculov@biocenter.kz)

Version: 2 Date: 10 Jul 2017

Author’s response to reviews:

We have made additions to the article according to the comments:

1. Please add a section "Additional files" - done

2. Please can you clarify in the Consent to publish section - done

3. Please add a “Conclusions” section after the “Discussion” section. - done

4. At this stage, please upload your manuscript (tables/additional files) as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. - done